Author's response to reviews

Title: Information and informed consent in infant vaccinations

Authors:

Francesco Attena (francesco.attena@unina2.it)
Amanda Valdes Abuadili (amanda.valdes@unina2.it)
Sara Marino (sara.marino@unina2.it)

Version: 2 Date: 1 August 2013

Author's response to reviews:

BMC Public Health
01/08/2013

Dear Editor,

Please find enclosed the revised version of our manuscript entitled “Infant Vaccination: Information and Informed Consent”, which we would like to submit for publication in BMC Public Health.

According to your suggestions (email 1.8.2013; MS:2695535761044527) we made the following changes in the Methods section:

1. The approval of this study was obtained by the Ethics Committee of the Second University of Naples.
2. A written informed consent has been obtained from each participant.

We confirm that this manuscript has not been published elsewhere and is not under consideration by another journal. All authors have approved the manuscript and agree with submission to BMC Public Health. The study did not receive any financial support.

Francesco Attena, Professore Ordinario
Department of Experimental Medicine
Second University of Naples
Via Luciano Armanni, 5
80138 Naples (Italy)
Tel & fax: +39 081 5666030
e-mail: francesco.attena@unina2.it

We look forward to hearing from you at your earliest convenience.

Yours sincerely,
Francesco Attena